Author's response to reviews

Title: Opinion versus practice regarding the use of rehabilitation services in home care: An investigation using machine learning algorithms

Authors:

Lu Cheng (l27cheng@uwaterloo.ca)
Mu Zhu (m3zhu@uwaterloo.ca)
Jeffrey W Poss (jwposs@uwaterloo.ca)
John P Hirdes (hirdes@uwaterloo.ca)
Christine Glenny (christine.glenny@gmail.com)
Paul Stolee (stolee@uwaterloo.ca)

Version: 2
Date: 20 February 2015

Author's response to reviews: see over
Editors
*BMC Medical Informatics and Decision Making*

February 20, 2015

Please accept the following revisions

1. Abstract – added in key words
2. Added in Authors’ Contribution
3. Added in Acknowledgements

Yours sincerely,

[Signature]

Paul Stolee, PhD
Associate Professor